Author's response to reviews

Title: Promoting advance planning for health care and research among older adults: a randomized controlled trial

Authors:

Gina Bravo (Gina.Bravo@USherbrooke.ca)
Marcel Arcand (Marcel.Arcand@USherbrooke.ca)
Danièle Blanchette (Daniele.Blanchette@USherbrooke.ca)
Anne-Marie Boire-Lavigne (Anne-Marie.Boire-Lavigne@USherbrooke.ca)
Marie-France Dubois (Marie-France.Dubois@USherbrooke.ca)
Maryse Guay (Maryse.Guay@USherbrooke.ca)
Paule Hottin (Paule.Hottin@USherbrooke.ca)
Julie Lane (Julie.Lane@USherbrooke.ca)
Judith Lauzon (judithl@videotron.ca)
Suzanne Bellemare (bellemaresuzanne@videotron.ca)

Version: 2 Date: 23 November 2011

Author's response to reviews: see over
Editor
BMC Medical Ethics

Re: Resubmission of Manuscript # 1650860652606205

Dear Editor:

I am submitting a revised manuscript that, I believe, has been formatted as requested. Required changes are listed below, followed by my responses (in bleu).

Please let me know if further changes are needed.

Sincerely,

Gina Bravo, Ph.D.
Principal investigator and corresponding author

**Major revision**

Title: please ensure that the title is identical in the manuscript and the submission system. **Done**

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. For example, in the manuscript one author is named as Danièle Blanchette, where as in the submission system they are named as Daniele Blanchette. **Danièle's name was corrected in the submission system.**

Please add the text '/design' to the 'methods' to make a 'methods/design' heading in the abstract. **Done**

**Minor revisions**

Title Page, Email address: please include email addresses for all the authors on the title page. **Email addresses were added on the title page.**

Tables: We note that you have included a table (with the legend 'Power to detect the intervention effect on documented preferences...') within the manuscript. Please place this table after the reference list alongside it's title and legend. **Done.**

Please remember that every table provided at the end of the manuscript file must be referred to in the text. **It is.**

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. **Here, I hope I have done what was requested. Not entirely sure.**

Please ensure that all the URL links work. **All URL links were checked.**

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. **Done. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. OK**